Author’s response to reviews

Title: Advice or Exercise for Chronic Whiplash Disorders? Design of a randomized controlled trial.

Authors:

Mark J Stewart (Mark.Stewart@fhs.usyd.edu.au)
Dr Christopher G Maher (C.Maher@fhs.usyd.edu.au)
Kathryn M Refshauge (K.Refshauge@fhs.usyd.edu.au)
Robert D Herbert (R.Herbert@fhs.usyd.edu.au)
Nikolai Bogduk (mgillam@mail.newcastle.edu.au)
Michael Nicholas (miken@mail.med.usyd.edu.au)

Version: 2 Date: 23 May 2003

As instructed, I have re-structured the abstract to reflect the structure of the main text. An additional Figure 1 was removed.